Author’s response to reviews

Title: Extraperitoneal laparoscopic resection for retroperitoneal lymphatic cyst: initial experience

Authors:

Yichun Wang (wycwangyichun@163.com)
Chen Chen (604942611@qq.com)
Chuanjie Zhang (ycwang1992@foxmail.com)
Chao Qin (urology_qinchao@163.com)
Ninghong Song (songninghongauthor@163.com)

Version: 1 Date: 14 Dec 2016

Author’s response to reviews:

Dear Editor

Thank you for your assessment of our manuscript submission. We had carefully considered your queries.

1. Consent to participate
This is a retrospective study about the laparoscopic retroperitoneal lymphocele resection. About the written informed consent mentioned in our article, it is a routine preoperative written informed consent. And we had just corrected the expression of this information.

2. Consent to publish
We misunderstood the meaning of consent to publication when we submitted the manuscript. It is different from consent to participate. And we add the consent to publication in the declaration. In our study written informed consent for publication of their clinical details and clinical images were obtained from all the patients.

Best wishes
Chao Qin